Author's response to reviews

Title: Design of cohort studies in chronic diseases using routinely collected databases when a prescription is used as surrogate outcome

Authors:

Sara Lodi (sal@ctu.mrc.ac.uk)
James Carpenter (james.carpenter@lshtm.ac.uk)
Peter Egger (peter.j.egger@gsk.com)
Stephen Evans (stephen.evans@lshtm.ac.uk)

Version: 5 Date: 4 March 2011

Author's response to reviews:

Dear Editor,

we have addressed all the major formatting revisions and when possible some of the minor ones.

Kind regards
Sara Lodi